Author's response to reviews

Title: Prognosis of screen-detected breast cancers: results of a population based study

Authors:

Laura Cortesi (hbc@unimo.it)
Vincenzo E Chiuri (chiuriv@unimore.it)
Silvia Ruscelli (silvia_ruscelli@hotmail.it)
Valeria Bellelli (v.bellelli@ausl.mo.it)
Rossella Negri (r.negri@ausl.mo.it)
Ivan Rashid (rashid.ivan@policlinico.mo.it)
Claudia Cirilli (cirilli.claudia@policlinico.mo.it)
Antonella Fracca (fracca.antonella@policlinico.mo.it)
Ennio Gallo (gallo.ennio@unimo.it)
Massimo Federico (federico@unimore.it)

Version: 8 Date: 23 December 2005

Author's response to reviews:

The Title and abstract were identical in the manuscript and submission system
The author Vincenzo Emanuele Chiuri has been changed in the manuscript as Vincenzo E Chiuri
The reference list contain all the authors' names and et al was eliminated
The manuscript has been edited according to the EndNote record